Author's response to reviews

Title: Effect of foot orthoses on lower extremity kinetics during running: a systematic literature review.

Authors:

Andrew McMillan (am3mcmillan-sexton@students.latrobe.edu.au)
Craig Payne (C.Payne@latrobe.edu.au)

Version: 4 Date: 14 November 2008

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

Please find attached a revised version of the manuscript with the following changes as requested:

Keywords on the title page have been removed.

The sub-title 'review' has been replaced with the sub-title 'results' in both the abstract and main text. This change was requested by the Journal of Foot and Ankle Research Editorial Team.

In-text citations of Tables 2 and 3 have been adjusted to appear in consecutive numerical order.

All vertical elements in Tables have been removed.

Thankyou for your support.

Andrew McMillan